Author’s response to reviews

Title: Combined use of high doses of vasopressin and corticosteroids in chron's disease patient with refractory septic shock after intestinal perforation: a case report

Authors:
Salvatore Notaro (notarosal@gmail.com)
Marcello Sorrentino (marcello.sorrentino@tiscali.it)
Aniello Ruocco (achille1979@hotmail.it)
Annalisa Notaro (annalisa.notaro@hotmail.it)
Antonio Corcione (corcio.ant@libero.it)
Patrizia Murino (patrizia.murino@ospedalemonaldi.it)
Eugenio Piscitelli (Eugenio.Piscitelli@libero.it)
Marianna Tamborino (mariannatamborino@libero.it)

Version: 2 Date: 03 Jul 2017

Author’s response to reviews:

Dear Pavel Michalek

I followed your advice and I did review article from language experts as suggested by you

thanks for the advice

Pending revaluation

I hope it may have been useful

Yours sincerely

Salvatore